Author's response to reviews

Title: Objective Structured Clinical Examinations (OSCEs), Psychiatry and the Clinical Assessment of Skills and Competencies (CASC). Same evidence, different judgement

Authors:

Steven Marwaha (s.marwaha@warwick.ac.uk)

Version: 2 Date: 20 April 2011

Author's response to reviews:

Dr Steven Marwaha
Associate Clinical Professor of Psychiatry
Rm 149, Medical School Building
Gibbet Hill Campus
Warwick Medical School
Coventry
CV47AL

20th April 2011

Editor
BMC Psychiatry

Dear Editor,

Re: OSCEs, Psychiatry and the CASC. Same evidence, different judgement

Thank you for considering this manuscript for publication in BMC Psychiatry and for the reviewer’s comments.

In response to these:

1. The reviewer has questioned the use of an abbreviation in the title. I note the editor has also asked that OSCE and CASC be defined in the title. I have now completed this as requested.

2. The reviewer comments “the use of the first person is a matter of taste”. I have now changed the structure of this sentence.

3. The reviewer asks for a correction to one of the references. I have now made this correction.

I have also corrected a number of typos and have marked any changes to the manuscript using the Track changes function. I hope these amendments meet
with your approval.

Yours sincerely

Dr Steven Marwaha